\documentstyle[2up]{article} 
\begin{document} 
\section{AdaptiveLoad.bib}
\cite{Zho} Zho\\ 
\cite{Vorn??} Vorn??\\ 
\cite{reiher-jefferson} reiher-jefferson\\ 
\cite{kurose-chipalkatti87} kurose-chipalkatti87\\ 
\cite{Kre92} Kre92\\ 
\cite{Chow82} Chow82\\ 
\cite{Hu61} Hu61\\ 
\cite{Jack63} Jack63\\ 
\cite{Gord67} Gord67\\ 
\cite{Klei67} Klei67\\ 
\cite{Coff68} Coff68\\ 
\cite{Denn86} Denn86\\ 
\cite{Grah69} Grah69\\ 
\cite{Wolf70a} Wolf70a\\ 
\cite{Wolf70b} Wolf70b\\ 
\cite{Coff72} Coff72\\ 
\cite{Buze73} Buze73\\ 
\cite{Frat73} Frat73\\ 
\cite{Adam74} Adam74\\ 
\cite{Labe74} Labe74\\ 
\cite{Nare74} Nare74\\ 
\cite{Ashc75} Ashc75\\ 
\cite{Bask75} Bask75\\ 
\cite{Ullm75} Ullm75\\ 
\cite{Zout76} Zout76\\ 
\cite{Chow77} Chow77\\ 
\cite{Gall77} Gall77\\ 
\cite{Gonz77a} Gonz77a\\ 
\cite{Gonz77b} Gonz77b\\ 
\cite{Rusc77} Rusc77\\ 
\cite{Ston77} Ston77\\ 
\cite{stone77} stone77\\ 
\cite{Gare78} Gare78\\ 
\cite{Koba78} Koba78\\ 
\cite{Rusc78} Rusc78\\ 
\cite{Solo78} Solo78\\ 
\cite{Ston78a} Ston78a\\ 
\cite{Ston78b} Ston78b\\ 
\cite{stone78} stone78\\ 
\cite{Webe78} Webe78\\ 
\cite{Birt79} Birt79\\ 
\cite{Bokh79} Bokh79\\ 
\cite{bokhari79} bokhari79\\ 
\cite{Chow79} Chow79\\ 
\cite{Grah79} Grah79\\ 
\cite{unknown1} unknown1\\ 
\cite{chou-kohler79} chou-kohler79\\ 
\cite{chu-holloway-lan-efe80} chu-holloway-lan-efe80\\ 
\cite{Chu80} Chu80\\ 
\cite{Coff80} Coff80\\ 
\cite{Gost80} Gost80\\ 
\cite{Ham80} Ham80\\ 
\cite{Hammerstrom80} Hammerstrom80\\ 
\cite{kratzer-hammerstrom80} kratzer-hammerstrom80\\ 
\cite{Oust80} Oust80\\ 
\cite{Reis80} Reis80\\ 
\cite{unknown2} unknown2\\ 
\cite{unknown6} unknown6\\ 
\cite{Wu80} Wu80\\ 
\cite{Yang90} Yang90\\ 
\cite{Bokh81a} Bokh81a\\ 
\cite{Brya81} Brya81\\ 
\cite{bryant-finkel81} bryant-finkel81\\ 
\cite{Case81} Case81\\ 
\cite{Klei81} Klei81\\ 
\cite{Lo81} Lo81\\ 
\cite{marcogliese-movarese81} marcogliese-movarese81\\ 
\cite{Neut81} Neut81\\ 
\cite{Ni81} Ni81\\ 
\cite{Pope81} Pope81\\ 
\cite{Tenn81a} Tenn81a\\ 
\cite{Tenn81b} Tenn81b\\ 
\cite{tilborg-wittie81b} tilborg-wittie81b\\ 
\cite{Agra82} Agra82\\ 
\cite{Andr82} Andr82\\ 
\cite{chou-abraham82} chou-abraham82\\ 
\cite{chou82} chou82\\ 
\cite{Cohe82} Cohe82\\ 
\cite{Davi82} Davi82\\ 
\cite{Doty82} Doty82\\ 
\cite{Efe82} Efe82\\ 
\cite{Heid82} Heid82\\ 
\cite{Hwan82} Hwan82\\ 
\cite{Livn82} Livn82\\ 
\cite{livny-melman82} livny-melman82\\ 
\cite{Livny82} Livny82\\ 
\cite{Ma82} Ma82\\ 
\cite{Mitr82} Mitr82\\ 
\cite{Oust82} Oust82\\ 
\cite{Zaho82} Zaho82\\ 
\cite{Davi83} Davi83\\ 
\cite{Gold83} Gold83\\ 
\cite{Koba83} Koba83\\ 
\cite{Krus83} Krus83\\ 
\cite{Mars83} Mars83\\ 
\cite{Powe83} Powe83\\ 
\cite{Barak84} Barak84\\ 
\cite{Berman84} Berman84\\ 
\cite{Cher84} Cher84\\ 
\cite{Chu84} Chu84\\ 
\cite{Dall84} Dall84\\ 
\cite{Gao84} Gao84\\ 
\cite{Henr84} Henr84\\ 
\cite{Kasa84} Kasa84\\ 
\cite{Kram84} Kram84\\ 
\cite{Kreuger84} Kreuger84\\ 
\cite{Lo84} Lo84\\ 
\cite{Mull84} Mull84\\ 
\cite{Rama84} Rama84\\ 
\cite{ramamritham-stankovic84} ramamritham-stankovic84\\ 
\cite{Schm84} Schm84\\ 
\cite{Souz84} Souz84\\ 
\cite{Stan84a} Stan84a\\ 
\cite{Stan84b} Stan84b\\ 
\cite{Stan84c} Stan84c\\ 
\cite{stankovic-sidhu84} stankovic-sidhu84\\ 
\cite{stankovic84} stankovic84\\ 
\cite{Tant84} Tant84\\ 
\cite{Til84} Til84\\ 
\cite{Yum84} Yum84\\ 
\cite{agrawal-ezzat85} agrawal-ezzat85\\ 
\cite{Bara85a} Bara85a\\ 
\cite{barak-shiloh85} barak-shiloh85\\ 
\cite{Berm84} Berm84\\ 
\cite{Berm85} Berm85\\ 
\cite{Care85} Care85\\ 
\cite{cheriton85} cheriton85\\ 
\cite{ezzat-agrawal85} ezzat-agrawal85\\ 
\cite{gait85} gait85\\ 
\cite{Lo85} Lo85\\ 
\cite{Mull85} Mull85\\ 
\cite{ni-hwang85} ni-hwang85\\ 
\cite{ni-xu-gendreau85} ni-xu-gendreau85\\ 
\cite{Ni85a} Ni85a\\ 
\cite{Ni85b} Ni85b\\ 
\cite{Peac84} Peac84\\ 
\cite{Pete85} Pete85\\ 
\cite{Sta85} Sta85\\ 
\cite{Stan85a} Stan85a\\ 
\cite{Stan85b} Stan85b\\ 
\cite{Stid85} Stid85\\ 
\cite{Tane85} Tane85\\ 
\cite{Tant85} Tant85\\ 
\cite{Tantawi85} Tantawi85\\ 
\cite{Thei85} Thei85\\ 
\cite{Vrsa85} Vrsa85\\ 
\cite{wang-morris85} wang-morris85\\ 
\cite{Wang85} Wang85\\ 
\cite{agrawal-jagadish86} agrawal-jagadish86\\ 
\cite{Ashr86} Ashr86\\ 
\cite{Bach86} Bach86\\ 
\cite{Casa86} Casa86\\ 
\cite{Chen86} Chen86\\ 
\cite{Dole86} Dole86\\ 
\cite{Eage86a} Eage86a\\ 
\cite{Eage86b} Eage86b\\ 
\cite{Eager86} Eager86\\ 
\cite{Ferr86} Ferr86\\ 
\cite{Ferrari86} Ferrari86\\ 
\cite{Gree86} Gree86\\ 
\cite{hac-jin86} hac-jin86\\ 
\cite{hac-johnson86} hac-johnson86\\ 
\cite{Hac86} Hac86\\ 
\cite{hagmann86a} hagmann86a\\ 
\cite{hagmann86b} hagmann86b\\ 
\cite{Haye86} Haye86\\ 
\cite{Iqbal86} Iqbal86\\ 
\cite{Kame86} Kame86\\ 
\cite{Lazo86} Lazo86\\ 
\cite{lee-towsley86} lee-towsley86\\ 
\cite{Lee86} Lee86\\ 
\cite{Lela86} Lela86\\ 
\cite{leland-ott86a} leland-ott86a\\ 
\cite{leland-ott86b} leland-ott86b\\ 
\cite{Leland86} Leland86\\ 
\cite{liu-silvester86} liu-silvester86\\ 
\cite{manber86} manber86\\ 
\cite{Mirc86} Mirc86\\ 
\cite{nachbar86} nachbar86\\ 
\cite{Nain86} Nain86\\ 
\cite{Pope86} Pope86\\ 
\cite{Rash86} Rash86\\ 
\cite{Tows86} Tows86\\ 
\cite{Wilf86} Wilf86\\ 
\cite{Wolf86} Wolf86\\ 
\cite{Yin86} Yin86\\ 
\cite{zhou86} zhou86\\ 
\cite{Berg87} Berg87\\ 
\cite{Blai87} Blai87\\ 
\cite{Bokh87} Bokh87\\ 
\cite{Cas87} Cas87\\ 
\cite{Casa87} Casa87\\ 
\cite{Chen87} Chen87\\ 
\cite{Chu87} Chu87\\ 
\cite{Cytr87} Cytr87\\ 
\cite{douglis-outserhout87} douglis-outserhout87\\ 
\cite{Hac87} Hac87\\ 
\cite{Hoss87} Hoss87\\ 
\cite{Hous87} Hous87\\ 
\cite{Kreuger87} Kreuger87\\ 
\cite{Krua87} Krua87\\ 
\cite{Krue87} Krue87\\ 
\cite{Lee87a} Lee87a\\ 
\cite{lee87b} lee87b\\ 
\cite{Lin87a} Lin87a\\ 
\cite{Lin87b} Lin87b\\ 
\cite{litzkow87} litzkow87\\ 
\cite{malone-fikes-grant-howard87} malone-fikes-grant-howard87\\ 
\cite{Mitr87} Mitr87\\ 
\cite{Mutk87a} Mutk87a\\ 
\cite{Mutk87b} Mutk87b\\ 
\cite{Papa87} Papa87\\ 
\cite{Pase87a} Pase87a\\ 
\cite{Reed87} Reed87\\ 
\cite{Ripl87} Ripl87\\ 
\cite{Sadayappan87a} Sadayappan87a\\ 
\cite{Sadayappan87b} Sadayappan87b\\ 
\cite{Sham87} Sham87\\ 
\cite{Shamir87} Shamir87\\ 
\cite{sinclair87} sinclair87\\ 
\cite{SIS87} SIS87\\ 
\cite{Sven87} Sven87\\ 
\cite{Thom87} Thom87\\ 
\cite{Vornberger87} Vornberger87\\ 
\cite{Zaya87} Zaya87\\ 
\cite{Zhou87a} Zhou87a\\ 
\cite{Zhou87b} Zhou87b\\ 
\cite{abbott-etal88} abbott-etal88\\ 
\cite{alonso-cova88} alonso-cova88\\ 
\cite{Bakk88} Bakk88\\ 
\cite{Baum88} Baum88\\ 
\cite{Baumgartner88} Baumgartner88\\ 
\cite{Berman88} Berman88\\ 
\cite{boel-schuppen88} boel-schuppen88\\ 
\cite{Bono88} Bono88\\ 
\cite{bonomi-kumar88b} bonomi-kumar88b\\ 
\cite{Call88} Call88\\ 
\cite{Casa88} Casa88\\ 
\cite{chen-reed88} chen-reed88\\ 
\cite{Chen88a} Chen88a\\ 
\cite{Chen88b} Chen88b\\ 
\cite{Cher88} Cher88\\ 
\cite{Du88} Du88\\ 
\cite{Dutt88} Dutt88\\ 
\cite{Eager88} Eager88\\ 
\cite{Ferguson88} Ferguson88\\ 
\cite{garcia-molina-kogan88} garcia-molina-kogan88\\ 
\cite{hac-jin88} hac-jin88\\ 
\cite{Hai88} Hai88\\ 
\cite{Hailperin88} Hailperin88\\ 
\cite{Kale88} Kale88\\ 
\cite{KayL88} KayL88\\ 
\cite{Lee88a} Lee88a\\ 
\cite{Lee88b} Lee88b\\ 
\cite{Leff88} Leff88\\ 
\cite{Lisk88} Lisk88\\ 
\cite{Litz88} Litz88\\ 
\cite{liu-silvester88} liu-silvester88\\ 
\cite{liu88} liu88\\ 
\cite{Lo88} Lo88\\ 
\cite{Lo88a} Lo88a\\ 
\cite{luk-wang-ling88} luk-wang-ling88\\ 
\cite{lyle-lu88} lyle-lu88\\ 
\cite{malone-fikes-grant-howard88} malone-fikes-grant-howard88\\ 
\cite{milutinovic-crnkovic-houstis88} milutinovic-crnkovic-houstis88\\ 
\cite{Mur88} Mur88\\ 
\cite{Nicol88} Nicol88\\ 
\cite{Notk88} Notk88\\ 
\cite{outserhout88} outserhout88\\ 
\cite{Pasq88} Pasq88\\ 
\cite{Pol88} Pol88\\ 
\cite{pulidas-towsley-stankovic88} pulidas-towsley-stankovic88\\ 
\cite{Rene88} Rene88\\ 
\cite{Romm88} Romm88\\ 
\cite{Ross88} Ross88\\ 
\cite{Shaw88} Shaw88\\ 
\cite{She88a} She88a\\ 
\cite{shoja-clarke-taylor-taylor88} shoja-clarke-taylor-taylor88\\ 
\cite{shoja-taylor88} shoja-taylor88\\ 
\cite{Smit88} Smit88\\ 
\cite{Souz88} Souz88\\ 
\cite{stumm88} stumm88\\ 
\cite{theimer-lantz88} theimer-lantz88\\ 
\cite{Wein88} Wein88\\ 
\cite{weinrib-shenker88} weinrib-shenker88\\ 
\cite{zhou88} zhou88\\ 
\cite{Arts89} Arts89\\ 
\cite{Baal89} Baal89\\ 
\cite{Bal89} Bal89\\ 
\cite{baumgartner-wah89} baumgartner-wah89\\ 
\cite{Baumgartner89} Baumgartner89\\ 
\cite{Bett89} Bett89\\ 
\cite{bodnar-liu89} bodnar-liu89\\ 
\cite{boel-schuppen89} boel-schuppen89\\ 
\cite{Cyb89} Cyb89\\ 
\cite{Cybe89} Cybe89\\ 
\cite{devarakonda-iyer89} devarakonda-iyer89\\ 
\cite{Doug91} Doug91\\ 
\cite{Dragon89} Dragon89\\ 
\cite{efe-groselj89} efe-groselj89\\ 
\cite{Efe89} Efe89\\ 
\cite{Ezzat86} Ezzat86\\ 
\cite{Gosw89} Gosw89\\ 
\cite{goswami-iyer-devarakonda89} goswami-iyer-devarakonda89\\ 
\cite{Goswami89} Goswami89\\ 
\cite{Hac89a} Hac89a\\ 
\cite{Hac89b} Hac89b\\ 
\cite{hailperin89} hailperin89\\ 
\cite{Hoss89} Hoss89\\ 
\cite{Hwan89} Hwan89\\ 
\cite{Jacq89} Jacq89\\ 
\cite{Kim89} Kim89\\ 
\cite{Kum89} Kum89\\ 
\cite{Leff89} Leff89\\ 
\cite{middleton89} middleton89\\ 
\cite{Mirc89} Mirc89\\ 
\cite{mirchandaney-towsley-stankovic89a} mirchandaney-towsley-stankovic89a\\ 
\cite{Motw89} Motw89\\ 
\cite{Rama89} Rama89\\ 
\cite{Ravi89} Ravi89\\ 
\cite{ravi@iag.hp.com}} ravi@iag.hp.com}\\ 
\cite{Ross89} Ross89\\ 
\cite{ryou-wong89} ryou-wong89\\ 
\cite{shenker-weinrib89} shenker-weinrib89\\ 
\cite{shin-chang89} shin-chang89\\ 
\cite{Shu89} Shu89\\ 
\cite{Vempaty89} Vempaty89\\ 
\cite{vranes89} vranes89\\ 
\cite{Will89} Will89\\ 
\cite{Willebeek-LeMa89} Willebeek-LeMa89\\ 
\cite{yan-lundstrom89} yan-lundstrom89\\ 
\cite{Ahuj90} Ahuj90\\ 
\cite{Alle90} Alle90\\ 
\cite{Bacc90} Bacc90\\ 
\cite{Blac90a} Blac90a\\ 
\cite{Black90b} Black90b\\ 
\cite{Burn90} Burn90\\ 
\cite{Cha90} Cha90\\ 
\cite{Chan90} Chan90\\ 
\cite{Chen90a} Chen90a\\ 
\cite{Chen90b} Chen90b\\ 
\cite{Cho80} Cho80\\ 
\cite{Chow90} Chow90\\ 
\cite{Chowdhury90} Chowdhury90\\ 
\cite{Dinn90} Dinn90\\ 
\cite{Elre90a} Elre90a\\ 
\cite{Elre90b} Elre90b\\ 
\cite{Elre90c} Elre90c\\ 
\cite{Erca90} Erca90\\ 
\cite{Eski90} Eski90\\ 
\cite{Essi90} Essi90\\ 
\cite{Feit90} Feit90\\ 
\cite{Fur90} Fur90\\ 
\cite{Gosc90} Gosc90\\ 
\cite{Gree90} Gree90\\ 
\cite{Grun90} Grun90\\ 
\cite{Hac90} Hac90\\ 
\cite{Han90} Han90\\ 
\cite{Hinz90} Hinz90\\ 
\cite{Hoss90} Hoss90\\ 
\cite{Hous90} Hous90\\ 
\cite{Jans90} Jans90\\ 
\cite{Jun90} Jun90\\ 
\cite{Kaas90} Kaas90\\ 
\cite{Lee90a} Lee90a\\ 
\cite{Lee90b} Lee90b\\ 
\cite{Leut90} Leut90\\ 
\cite{Leve90} Leve90\\ 
\cite{Li90} Li90\\ 
\cite{Mirc90} Mirc90\\ 
\cite{Mull90} Mull90\\ 
\cite{nicol-reynolds90} nicol-reynolds90\\ 
\cite{nicol-saltz90} nicol-saltz90\\ 
\cite{Rot90} Rot90\\ 
\cite{Saletore90} Saletore90\\ 
\cite{Shi90b} Shi90b\\ 
\cite{Shir90} Shir90\\ 
\cite{Shiv90} Shiv90\\ 
\cite{Suhl90} Suhl90\\ 
\cite{Tane90} Tane90\\ 
\cite{Woli90} Woli90\\ 
\cite{Yua90} Yua90\\ 
\cite{Zaho90} Zaho90\\ 
\cite{Abra91} Abra91\\ 
\cite{Ahm91} Ahm91\\ 
\cite{Ahma82} Ahma82\\ 
\cite{Andr91} Andr91\\ 
\cite{Ange90} Ange90\\ 
\cite{Aylo91} Aylo91\\ 
\cite{Baden91} Baden91\\ 
\cite{Bern71} Bern71\\ 
\cite{Blac91} Blac91\\ 
\cite{Bokh81b} Bokh81b\\ 
\cite{Brau91} Brau91\\ 
\cite{Cerf91} Cerf91\\ 
\cite{Chan91} Chan91\\ 
\cite{Chen91} Chen91\\ 
\cite{Dasg91} Dasg91\\ 
\cite{DeKeyser91} DeKeyser91\\ 
\cite{Dert91} Dert91\\ 
\cite{Duch91} Duch91\\ 
\cite{Elre91} Elre91\\ 
\cite{Eski91} Eski91\\ 
\cite{Fidg91} Fidg91\\ 
\cite{gopinath-gupta91} gopinath-gupta91\\ 
\cite{Gosc91} Gosc91\\ 
\cite{Goud91} Goud91\\ 
\cite{Gray91} Gray91\\ 
\cite{Gree91} Gree91\\ 
\cite{grimshaw-vivas91} grimshaw-vivas91\\ 
\cite{Hald91} Hald91\\ 
\cite{Hanx91} Hanx91\\ 
\cite{Hari91} Hari91\\ 
\cite{Hell91} Hell91\\ 
\cite{Henn91} Henn91\\ 
\cite{Hous91} Hous91\\ 
\cite{Hwan91} Hwan91\\ 
\cite{jacqmot-milgrom91} jacqmot-milgrom91\\ 
\cite{Kaas91} Kaas91\\ 
\cite{Kou91} Kou91\\ 
\cite{kreuger-chawla91} kreuger-chawla91\\ 
\cite{Kum91} Kum91\\ 
\cite{Leff91} Leff91\\ 
\cite{Lue91} Lue91\\ 
\cite{Mafl91} Mafl91\\ 
\cite{Marc91} Marc91\\ 
\cite{Mart91} Mart91\\ 
\cite{Marz91} Marz91\\ 
\cite{Moru91} Moru91\\ 
\cite{Nitz91} Nitz91\\ 
\cite{Romm91} Romm91\\ 
\cite{Rosi91} Rosi91\\ 
\cite{Ross91} Ross91\\ 
\cite{schaar-efe-delcambra-bhuyan91} schaar-efe-delcambra-bhuyan91\\ 
\cite{Sing91} Sing91\\ 
\cite{Sinh91} Sinh91\\ 
\cite{Smit91} Smit91\\ 
\cite{Souz91} Souz91\\ 
\cite{Ston90} Ston90\\ 
\cite{Stro91} Stro91\\ 
\cite{Tane91} Tane91\\ 
\cite{Tesl91} Tesl91\\ 
\cite{Tri91} Tri91\\ 
\cite{Vzsu91} Vzsu91\\ 
\cite{Weis91} Weis91\\ 
\cite{Williams91} Williams91\\ 
\cite{Witt91} Witt91\\ 
\cite{Woli91} Woli91\\ 
\cite{WilliamsCPE2} WilliamsCPE2\\ 
\cite{Ananth92} Ananth92\\ 
\cite{Ande92} Ande92\\ 
\cite{Biagioni92} Biagioni92\\ 
\cite{Bowe92} Bowe92\\ 
\cite{Clar92} Clar92\\ 
\cite{Conrad92} Conrad92\\ 
\cite{DeKeyser92} DeKeyser92\\ 
\cite{DeKeyserSHPCC7} DeKeyserSHPCC7\\ 
\cite{Epem92} Epem92\\ 
\cite{Gree92} Gree92\\ 
\cite{Haddad92} Haddad92\\ 
\cite{Lin92} Lin92\\ 
\cite{Lue92} Lue92\\ 
\cite{Mutk92} Mutk92\\ 
\cite{Nicol92} Nicol92\\ 
\cite{Nor92} Nor92\\ 
\cite{Schw92} Schw92\\ 
\cite{Shi92} Shi92\\ 
\cite{Sinha92} Sinha92\\ 
\cite{Sin93} Sin93\\ 
\cite{Chen??} Chen??\\ 
\cite{Gerr??} Gerr??\\ 
\bibliographystyle{unsrt}
\bibliography{AdaptiveLoad}
\end{document} 
